\documentclass{article}
\input{6045preamble}


\begin{document}
\homework{2}{Prof. Ron Rivest}{Due: 20 February 2002}{}

\begin{problem}
  Problem 1.4 (page 84) parts a, b, g, i, j  
\end{problem}

\begin{problem}
  Problem 1.6, part a. Do this problem twice, once producing a NFA via
  the construction of Theorem 1.22, and again producing a DFA via the
  cross-product construction from class.
\end{problem}

\begin{problem}
  Problem 1.7, part a
\end{problem}

\begin{problem}
  Problem 1.8 (page 85), part b
\end{problem}


\begin{problem}
Problem 1.12 
\end{problem}


\begin{problem}
  Problem 1.31 (page 89)
\end{problem}

\end{document}

%%% Local Variables: 
%%% mode: latex
%%% TeX-master: t
%%% End: 
